Author's response to reviews

Title: Designing and implementing a social protection intervention to enhance TB control: operational evidence from the CRESIPT project in Lima, Peru

Authors:

Tom Wingfield (tomwingfield@hotmail.co.uk)
Delia Boccia (delia.boccia@lshtm.ac.uk)
Marco Tovar (marco.tovar@ipsyd.org)
Doug Huff (dhuff@tulane.edu)
Rosario Montoya (mrmontoyav@gmail.com)
James J Lewis (james.lewis@lshtm.ac.uk)
Carlton A Evans (carlton.evans@ifhad.org)

Version: 4
Date: 9 January 2015

Author's response to reviews:

Dear Editors,

Please see the attached cover letter dated 9th January 2015.

Kind regards,

Tom Wingfield (on behalf of the authors)